

Citation: Lugo-Martinez J, Pejaver V, Pagel KA, Jain S, Mort M, Cooper DN, et al. (2016) The Loss and Gain of Functional Amino Acid Residues Is a Common Mechanism Causing Human Inherited Disease. PLoS Comput Biol 12(8): e1005091. doi:10.1371/journal.pcbi.1005091

Editor: Yana Bromberg, Rutgers University, UNITED STATES

Received: July 6, 2015

Accepted: August 2, 2016

Published: August 26, 2016

Copyright: © 2016 Lugo-Martinez et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: HGMD Professional is commercial and can be obtained through the license with Cardiff University. Additionally, there is a public version of HGMD data which validates the conclusions on this manuscript available at http://www.hgmd.cf.ac.uk/docs/register.html. All other data are available at the following link: http://www.cs.indiana.edu/~predrag/data/lugomartinez_ploscb_2016_datasets.zip.

Funding: JLM was supported by the Ford Foundation pre-doctoral fellowship, DNC and MM are grateful to Qiagen Inc for their financial support **RESEARCH ARTICLE**

The Loss and Gain of Functional Amino Acid Residues Is a Common Mechanism Causing Human Inherited Disease

Jose Lugo-Martinez¹, Vikas Pejaver¹, Kymberleigh A. Pagel¹, Shantanu Jain¹, Matthew Mort², David N. Cooper², Sean D. Mooney³*, Predrag Radivojac¹*

 Department of Computer Science and Informatics, Indiana University, Bloomington, Indiana, United States of America, 2 Institute of Medical Genetics, School of Medicine, Cardiff University, Cardiff, United Kingdom, 3 Department of Biomedical Informatics and Medical Education, University of Washington, Seattle, Washington, United States of America

* sdmooney@uw.edu (SDM); predrag@indiana.edu (PR)

Abstract

Elucidating the precise molecular events altered by disease-causing genetic variants represents a major challenge in translational bioinformatics. To this end, many studies have investigated the structural and functional impact of amino acid substitutions. Most of these studies were however limited in scope to either individual molecular functions or were concerned with functional effects (e.g. deleterious vs. neutral) without specifically considering possible molecular alterations. The recent growth of structural, molecular and genetic data presents an opportunity for more comprehensive studies to consider the structural environment of a residue of interest, to hypothesize specific molecular effects of sequence variants and to statistically associate these effects with genetic disease. In this study, we analyzed data sets of disease-causing and putatively neutral human variants mapped to protein 3D structures as part of a systematic study of the loss and gain of various types of functional attribute potentially underlying pathogenic molecular alterations. We first propose a formal model to assess probabilistically function-impacting variants. We then develop an array of structure-based functional residue predictors, evaluate their performance, and use them to quantify the impact of disease-causing amino acid substitutions on catalytic activity, metal binding, macromolecular binding, ligand binding, allosteric regulation and post-translational modifications. We show that our methodology generates actionable biological hypotheses for up to 41% of disease-causing genetic variants mapped to protein structures suggesting that it can be reliably used to guide experimental validation. Our results suggest that a significant fraction of disease-causing human variants mapping to protein structures are function-altering both in the presence and absence of stability disruption.

through a License Agreement with Cardiff University. We gratefully acknowledge the support from the National Institutes of Health through the awards R01LM009722 and R01MH105524. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

Author Summary

Identifying the molecular changes caused by mutations is a major challenge in understanding and treating human genetic disease. To address this problem, we have developed a wide range of profiling tools designed to predict specific types of functional site from protein 3D structures. We then apply these tools to data sets of inherited disease-associated and putatively neutral amino acid substitutions and estimate the relative contribution of the loss and gain of functional residues in disease. Our results suggest that alterations of molecular function are involved in a significant number of cases of human genetic disease and are over-represented as compared to putatively neutral variants. Additionally, we use experimental data to show that it is possible to computationally identify the loss of specific functional events in disease pathogenesis. Finally, our methodology can be used to reliably identify the potential molecular consequences of disease-causing genetic variants and hence prioritize experimental validation.

Introduction

Spurred by the advances in DNA sequencing, the accumulation of human genetic variation (and with it amino acid substitution data) has over the past two decades been unprecedented. Multiple databases and resources now enumerate and annotate amino acid substitutions, their functional impact, and association with inherited disease [1-3]. However, to further our understanding of human genetic variation and its impact on disease, it is necessary to elucidate the associated molecular alterations [4-6]. Thus, the step of identifying the underlying molecular mechanisms constitutes a serious impediment to understanding and treating human disease.

A straightforward approach to integrating genetic and molecular data is to search databases for structural and functional annotations at the variation site or in the neighborhood of interest, and then provide both the possible and the likely effects of mutations on these annotations [7-12]. Although this approach is useful, its major limitation is its dependence on previously observed and curated functional information as well as our inability, except in limited cases [9, 10], to cover mutations that create functional residues. Furthermore, the deterministic nature of data integration does not easily lend itself to a principled strategy of prioritizing many of the possible molecular mechanisms based on their likelihood to impact clinical phenotype, especially when a variant resides in the neighborhood of the functional site.

A more comprehensive approach to analyzing the effects of amino acid substitutions involves the use of statistical inference methods that predict functional impact. Although there are many studies that have adopted this strategy using the data from protein sequence or structure [13], most methods make inferences without specifying which functional property has been impacted. Such an approach, however, is feasible if the methodology can be developed to predict a specific function, say a phosphorylation site or a catalytic residue, which is then applied to sequence variants [14–16]. Furthermore, these specific functional predictions can be integrated with general variant effect predictors to provide probabilistic estimates of molecular mechanisms of disease [17].

While many successful machine learning models can be made based on sequence information alone, structural information can provide additional benefits [18, 19]. This suggests that further improvements could be made if specific predictors of protein function could be integrated into this pipeline [20]. Wang and Moult published a seminal work on the impact of germline variants on protein function [7]. They searched the Protein Data Bank (PDB) and used homology modeling to obtain 3D structures of wild-type proteins as a means to characterize the structural and functional effects of both disease and neutral variants. They reported that the majority of disease-causing substitutions affect protein stability, whereas a relatively small proportion directly disrupt molecular function. By contrast, Sahni et al. observed a rather larger fraction of function-impacting variants in their experimental studies into the impact of variants on protein-protein and protein-DNA interactions [21]. Their work therefore challenges the traditional view of the dominance of structure-impacting changes. Finally, Steward et al. examined the structural, functional and physicochemical features of wild-type protein structures where disease-causing variants occur [22]. Unfortunately, the scope of these and several other studies was limited to characterizing the functional effects of amino acid substitutions across a handful of protein functions [23, 24]. There is therefore a need for large-scale studies that use statistical inference methods based on protein structure to explore the relative contributions made by disruption of functional sites in disease pathogenesis.

In this work, we carry out a systematic study of the alterations of specific functional sites as the underlying molecular mechanisms of disease over a data set comprising germline diseasecausing amino acid substitutions mapped to protein 3D structures. In particular, we develop multiple structure-based functional residue predictors and assess the impact of disease-associated substitutions on catalytic residues, metal-binding sites, macromolecular binding sites, ligand-binding sites, allosteric sites and post-translational modification (PTM) sites. We then quantify the extent to which disruption or introduction of particular types of functional site accounts for the deleterious impact of amino acid substitutions. Our results provide evidence to support the view that the increased and decreased propensity of particular functional activities are common in human inherited disease.

Materials and Methods

Probabilistic model for alteration of residue function

For a given protein structure and a missense variant, we are interested in estimating whether a particular residue functionality, say *f*, has been impacted. To achieve this, we broadly distinguish between two scenarios resulting in alteration of function: (i) the mutation disrupts protein stability and subsequently impacts residue function and (ii) the mutation does not impact stability and structure yet still leads to an altered function as a consequence of modified functional propensity. The latter scenario might occur, for example, for tyrosine-to-phenylalanine substitutions that result in minimal structural changes, yet the mutation itself may have significant impact on protein phosphorylation and downstream events. This is because phenylalanine cannot be phosphorylated to subsequently create an SH2 binding site [25]. We informally refer to the events of increased or decreased functional propensity as gain and loss of functional activity.

We formalize this approach as follows. Let *x* be a collection of features that encodes a particular mutation in a protein structure and let P(loss of f|x) be the probability of the loss of residue function *f* consequent to mutation. Then, we can write

$$P(\text{loss of } f|x) = P(\text{loss of } f|S, x)P(S|x) + P(\text{loss of } f|S, x)P(S|x)$$

where the event *S* indicates that protein stability is significantly changed and *S* indicates that the stability is not significantly changed; i.e., $P(S|x) = 1 - P(\overline{S}|x)$. This expression gives a probabilistic formulation that can be used to estimate loss of a specific function *f*, with an assumption of a dichotomized impact on protein stability. We use this simple approach in part because of the issues involved in obtaining large amounts of high-quality data to assess the impact of sequence variants on stability. We note, however, that the expression can be generalized to multiple groups of stability disruption and in the limit the sum would turn into an integral. Now we briefly discuss estimating P(S|x), P(loss of f|S, x), and $P(\text{loss of } f|\overline{S}, x)$. The posterior probability of stability disruption P(S|x) can be determined by developing a computational model given a representative data set of variants that significantly impact stability of the protein (both stabilizing and destabilizing mutations) and a representative set of variants that do not. The negative data set can also be substituted by a large representative set of variants for which the impact on stability is unknown [26, 27]. This formulation falls into the category of positive-unlabeled learning [28], a version of semi-supervised learning in which the set of negative examples is unavailable or ignored; e.g., because available negative examples are biased.

Next, we discuss estimating $P(\text{loss of } f|\overline{S}, x)$; i.e., the probability of the loss of function given that there is no significant stability disruption. Let x' be a collection of features that encodes the structural environment of a residue and let f be the specific residue function of interest; e.g., whether the residue is phosphorylated, DNA-binding, etc. Let $P(f|x'_{wt})$ denote the probability that the residue at this site is functional in the wild-type protein and $P(f|x'_{mt})$ the probability that the mutated residue, at the same locus in the protein, is functional. We then define the probability of the loss of residue function f as

$$P(\text{loss of } f|\bar{S}, x) = P(f|x'_{wt}) \cdot (1 - P(f|x'_{mt})),$$

where $1 - P(f|\mathbf{x}'_{mt})$ gives the probability that the residue in the mutant protein is not functional. To estimate $P(\text{loss of } f|\overline{S}, x)$, we can employ the same functional residue predictor to compute prediction scores on the wild-type and mutant proteins. That is, because there are no structural changes, the same structure-based classifier can be used to compute both $P(f|\mathbf{x}'_{wt})$ and $P(f|\mathbf{x}'_{mt})$, with the only difference being the replaced amino acid in the feature set \mathbf{x}'_{mt} . As before, the probabilistic model $P(f|\mathbf{x}')$ can be developed using data sets of positive and negative examples or, in the absence of representative negative examples, using positive and unlabeled data. We will discuss the details of approximating the posterior probability of protein function in the next section. Finally, we can consider that large changes in protein stability and structure always abolish the function of a residue; i.e., $P(f|\mathbf{x}'_{mt}) \approx 0$. This implies that

$$P(\text{loss of } f|S, x) \approx P(f|x'_{wt}),$$

that is, the probability of the loss of function at a particular residue is roughly equal to the probability that the residue was functional in the first place.

In addition to the loss of protein function, we can also consider the event of the gain of residue function, where

$$P(\text{gain of } f|\bar{S}, x) = (1 - P(f|x_{wt})) \cdot P(f|x_{mt}).$$

This formulation accounts for the changes in residue microenvironments that increase its functional propensity. While most amino acid substitutions found in nature are neutral or disruptive, there are many examples in which they lead to the gain-of-function events. For example, generation of a sequence motif NX[S/T] has been observed to result in gain of N-linked glycosylation events and disease [9]. Similarly, changes in catalytic residues have been observed to increase the efficiency of catalysis, also with phenotypic implications [29]. Furthermore, assuming that significant stability changes rarely lead to gain of function we can simply take

$$P(\text{gain of } f|S, x) \approx 0.$$

In this paper, we are interested in specific types of residue function f for which sufficiently large data sets could be extracted from biological databases. We organize these residues into catalytic residues, metal-binding residues, macromolecule-binding residues, ligand-binding

residues, post-translationally modified sites, and allosteric residues. For the purposes of our study, we consider certain types of residues to be functional although they may also be important for protein stability. For example, a disruption of certain metal-binding sites, say a Zn²⁺-binding residue, will be considered here as disruption of functional residues that consequently impacts protein stability. We do, however, note that this distinction is somewhat philosophical.

Training stability predictors and functional residue predictors

All classification models in this work were trained using the positive-unlabeled framework in which we are given a set of positive examples and a set of unlabeled examples. In the case of stability predictors, P(S|x), the positive examples represent mutations that have been experimentally shown to significantly impact protein stability; based on the previous studies we selected these mutations to be either stabilizing or destabilizing with $|\Delta\Delta G| > 0.5$ kcal/mol [30], although some other studies use higher values [31, 32]. The set of unlabeled examples, on the other hand, was selected using a database of human variants, dbSNP, mapped to available protein structures in PDB. In the case of functional residue predictors P(f|x'), the positive examples were selected by integrating structural and molecular data that provide experimentally observed functional residues, whereas the unlabeled examples were selected from a set of monomeric proteins in PDB. We will describe all data sets precisely at the end of the Methods section.

We next discuss how to train a classification model from positive and unlabeled data. Let $\mathcal{D}_L = \{(x_i, y_i)\}_{i=1}^m$ be a labeled data set, where $x_i \in \mathcal{X}$ is an input example and $y_i \in \{-1, +1\}$ is its class label. Let $\mathcal{D}_U = \{x_i\}_{i=1}^n$ be a set of unlabeled examples. In the problem of learning whether a mutation impacts stability, x encodes a set of features corresponding to the mutation and y = +1 indicates large stability disruption. Similarly, in the case of functional site predictors, x encodes a particular residue microenvironment in a protein and y = +1 indicates that the residue is functional. In the positive-unlabeled formulation, all examples in \mathcal{D}_L have positive class labels, whereas \mathcal{D}_U is a mixture of positive and negative examples. The probability of positive examples P(y = +1) in the unlabeled set is referred to as the class prior. The task of the predictor is to learn the probability P(y = +1|x) when provided data sets \mathcal{D}_L and \mathcal{D}_U .

Unfortunately, learning P(y = +1|x) is not straightforward because the negative examples are not available. To address this problem we rely on the body of work in semi-supervised learning that decomposes the problem into the training of a non-traditional classifier [26]; i.e., a model that distinguishes between labeled and unlabeled data, and estimating the class prior P(y = +1). We denote the posterior probability from a non-traditional classifier as P(l = +1|x), where l = +1 refers to the event of data point being labeled. We approximate these probabilities using kernel-based learning with support vector machine (SVM) classifiers as underlying optimization engines. Additionally, we estimate P(y = +1) using the AlphaMax algorithm [27], and point out other available options for an interested reader [26, 33].

Under mild assumptions [27], the output of a non-traditional classifier P(l = +1|x) can be converted into the output of a traditional classifier P(y = +1|x) using

$$P(y = +1|x) = P(y = +1) \cdot \frac{n}{m} \cdot \frac{P(l = +1|x)}{1 - P(l = +1|x)},$$

where *m* and *n* are the sizes of labeled and unlabeled data sets, respectively. This predictor can now be applied to any data set \mathcal{D} to compute the frequency of the phenomenon using the empirical mean formula $\frac{1}{|\mathcal{D}|} \sum_{x \in \mathcal{D}} P(y = +1|x)$.

The probability of alteration for multiple types of function

We previously considered the loss and gain of the specific function f at a particular residue of interest. We now extend this definition to multiple types of functional residues as follows. Consider an event of loss of any function f from a set \mathcal{F} . We can use previous reasoning to re-write the earlier expression as

$$P(\text{loss of } \mathcal{F}|x) = P(\text{loss of } \mathcal{F}|S, x)P(S|x) + P(\text{loss of } \mathcal{F}|\bar{S}, x)P(\bar{S}|x).$$

To compute this probability, we need to compute probabilities $P(\text{loss of } \mathcal{F}|S, x)$ and $P(\text{loss of } \mathcal{F}|\overline{S}, x)$. Because the functional data is too sparse to learn the joint (posterior) models of residue function, we consider two models to approximate this probability using the marginal (posterior) models that the residue is functional. In the first model that we refer to as the independence model, we consider each type of functional residue to be independent of others and write

$$P(\mathrm{loss} ext{ of } \mathcal{F}|ar{S},x) = 1 - \prod_{f \in \mathcal{F}} (1 - P(\mathrm{loss} ext{ of } f|ar{S},x)).$$

The expression above is the probability that at least one of the functions from \mathcal{F} has been lost. Because the functions are not in reality independent, this model may lead to overestimation. The second, more conservative model, approximates the probability of loss as

$$P(\text{loss of } \mathcal{F}|\overline{S}, x) = \max_{f \in \mathcal{F}} \{P(\text{lossof} f|\overline{S}, x)\}.$$

We refer to this model as the max model. Equivalent expressions can be written for $P(\text{loss of } \mathcal{F}|S, x)$ as well as for the gain-of-function events. We note that \mathcal{F} may contain particular groups of functions, say all types of metal binding, or can be used for all functions considered in this work.

Graphlet kernels

In this section we briefly summarize the graphlet kernel prediction framework and show how these kernels were used to train both stability predictors and functional site predictors.

Graphs. A graph *G* is a pair (*V*, *E*), where *V* is a set of vertices (nodes) and $E \subseteq V \times V$ is a set of edges. In a vertex-labeled (colored) graph, a labeling function *g* is defined as $g: V \to \Sigma$, where Σ is a finite alphabet, commonly referred to as vertex alphabet. A graph without self-loops, i.e. where $(v, v) \notin E, \forall v \in V$, is said to be simple. An undirected graph is a graph where the order of the vertices in each pair $(u, v) \in E$ can be ignored; otherwise, the graph is said to be a directed graph. A rooted graph *G* is a graph together with a distinguished vertex termed the root.

Graphlets. A graphlet is a small, simple, connected, rooted graph. We refer to a graphlet with *n* vertices as an *n*-graphlet. For more information on graphlets, we direct the reader to [34-38].

Edit distance graphlet kernels. Consider a vertex-labeled graph $G = (V, E, g, \Sigma)$, where $|\Sigma| \ge 1$. Lugo-Martinez and Radivojac [38] defined the *m*-edit distance representation of vertex *v* as

$$\phi_{(n,m)}(v) = (\psi_{(n_1,m)}(v), \psi_{(n_2,m)}(v), \dots, \psi_{(n_{\kappa(n,\Sigma)},m)}(v)),$$

where

$$\psi_{(n_i,m)}(v) = \sum_{n_j \in E(n_i,m)} w(n_i,n_j) \cdot \varphi_{n_j}(v).$$

In the previous expression $\varphi_{n_j}(v)$ is the count of the *j*-th labeled *n*-graphlet rooted at $v, \kappa(n, \Sigma)$ is the total number of vertex-labeled *n*-graphlets and $E(n_i, m)$ is a set of *n*-graphlets such that for each $n_j \in E(n_i, m)$ there exists an edit distance path of length at most *m* that transforms n_i into n_j . That is, the number of edit operations necessary to transform n_i into n_j is at most *m*, where edit operations are defined as insertion or deletion of vertices and edges, or in the case of labeled graphs, substitutions of vertex and edge labels. Finally, weights $w(n_i, n_j) \ge 0$ are used to adjust the influence of pseudo counts and control computational complexity; in this study, we set w- $(n_i, n_j) = 1$ if $n_j \in E(n_i, m)$ and $w(n_i, n_j) = 0$ otherwise.

The length-*m* edit distance *n*-graphlet kernel $k_{(n,m)}(u, v)$ between vertices *u* and *v* can be computed as an inner product between the respective count vectors $\phi_{(n,m)}(u)$ and $\phi_{(n,m)}(v)$. Hence, the length-*m* edit distance graphlet kernel function can be expressed as

$$k_m(u,v) = \sum_{n=1}^N k_{(n,m)}(u,v),$$

where *N* is a small integer; typically defined up to N = 5 for undirected graphs. Additionally, one can define two subclasses of edit distance kernels referred to as (vertex) label-substitution k^l (only allows substitutions of vertex labels) and edge-indel kernels k^e (only allows insertion or deletion of edges). It is worth noting that if m = 0, then k_m , k_m^l and k_m^e are all equivalent to the standard graphlet kernel on labeled graphs [37]. In this work, we only considered the normalized kernel calculated as

$$k(u,v) = rac{k^*(u,v)}{\sqrt{k^*(u,u) \, k^*(v,v)}},$$

where $k^*(u, v)$ can be $k_m(u, v)$, $k_m^l(u, v)$, or $k_m^e(u, v)$. The normalized kernel has been previously shown to have favorable performance with respect to non-normalized kernels [37, 38].

Practical aspects of training. We used the graphlet kernel framework and SVM classifiers to construct all functional site predictors. First, we modeled protein structures as protein contact graphs, where each amino acid residue was represented as a vertex and two spatially close residues (i.e. 4.5Å or less between any two atoms) were linked by an undirected edge. Fig 1 illustrates a contact graph for a protein kinase rooted at a tyrosine residue at position 148. Next, we computed a set of normalized graphlet kernel matrices \mathcal{K} using $k_m^l(x_i, x_j)$, $k_m^e(x_i, x_j)$ and $k_m(x_i, x_j)$ for all pairs (x_i, x_j) . For each $k \in \mathcal{K}$, we used SVM^{light} [39] and the default value for the capacity parameter to train a predictor. We incorporated evolutionary information by extending the vertex alphabet Σ from the 20 standard amino acids to 40 based on the median residue conservation observed over the entire data set [38]. For example, the amino acid alanine was split into highly conserved alanines (represented as A) and other alanines (represented as a). Once each predictor was trained, we used Platt's correction to adjust the outputs of the predictor to the 0-1 range [40].

In the case of stability predictors, we augmented the graphlet kernel representation using 33 features previously shown to be informative [$\underline{41}$, $\underline{42}$]. In particular, we used 20 features for the 20 different amino acids to encode the mutation information (i.e. -1 for wild-type residue; +1 for mutated residue; 0 otherwise), and 13 features to encode the difference between physico-chemical properties between wild-type and mutant amino acid residues [$\underline{43}$ - $\underline{50}$].

Evaluation of in silico predictions

The performance of each predictor was first evaluated through a per-chain 10-fold cross-validation. In each iteration of cross-validation, 10% of protein chains were selected for the test set,





Fig 1. Left: Structure of human aurora kinase A, chain A fragment (PDB entry 2j4z) with highlighted phosphorylation site Tyr148 (denoted as Y148). Right: Corresponding level-3 protein contact graph centered at Tyr148 (denoted with double circles). Nodes represent amino acid residues and edges correspond to spatially neighboring residues (i.e. 6Å or less between C_{α} atoms).

doi:10.1371/journal.pcbi.1005091.g001

whereas the remaining 90% were used for training. This enforces that all data points from the same protein sequence belong to either training or test set and, thus, reduces the chance of overestimating the accuracy of the models. We estimated the area under the ROC curve (AUC), which plots the true positive rate as a function of the false positive rate and the Mat-thews correlation coefficient (MCC).

It is impractical to validate, in vitro or in vivo, the functional effects of each amino acid substitution in our data sets. Therefore, we used independent mutagenesis experimental data to additionally evaluate the performance of our functional site predictors and also evaluate predictions of the loss of functional residues. More specifically, we downloaded all human mutagenesis experimental data from UniProt as of September 2014. This data set comprised 14,933 substitutions from 3,044 distinct proteins. We removed all entries associated with more than one substitution. The resulting 11,425 sites were mapped to high-quality PDB structures using the same steps described in the next section. The final data set comprised 3,356 amino acid substitutions from 2,809 different sites in 880 proteins.

For each site in this data set, we extracted functional annotations related to metal binding, PTMs, active sites, macromolecular binding, ligand binding and allosteric activity. Then, for each functional site predictor, we built an independent test set such that (i) each site belonged to a chain that was less than 40% identical to any chain in the training data, (ii) there were at least five positive sites in each test set. The resulting set was used to assess the performance of the functional predictors independently of the cross-validation.

Similarly, we created a test data set to evaluate loss-of-function predictions as follows: we searched the description of the mutagenesis experiment such that there was an experimentally

Data set name	n _{AAS}	ns	n _{PDB}	n _c
Inherited disease	52406	10629	1177	1387
Unlabeled variants	282625	8417	3121	3585
Putatively neutral	282625	8049	3047	3500

Table 1. Summary	of amino acid substitution (AAS)	data sets.
------------------	------------------------------	------	------------

For each data set, we show the total number of amino acid substitutions (n_{AAS}), the number of substitutions mapped to PDB (n_s), the number of PDB entries (n_{PDB}) and the number of protein chains (n_c).

doi:10.1371/journal.pcbi.1005091.t001

observed disruption of a functional site. The resulting non-redundant and filtered data sets were then used to estimate the AUC and MCC of all loss-of-function predictors. We attempted to carry out the same steps for gain-of-function events but there were insufficient data.

Disease-associated mutation data

Missense variants causing inherited disease were obtained from the Human Gene Mutation Database (HGMD) as of June 2013. A set of unlabeled inherited variants was downloaded from dbSNP v.137. All amino acid substitutions were then mapped to protein structures in PDB as follows: (i) a database of amino acid sequences from X-ray crystallographic protein structures with more than 50 amino acids and resolution less than 2.5Å was created, and (ii) for each variant, a 51 residue long sequence centered around the wild-type amino acid at the variant position was aligned using BLAST [51] against the atom sequences in PDB. All alignments without an exact match; i.e., with gaps or sequence identity lower than 100%, were excluded from this study and in the case of multiple exact matches, the structure with the best resolution was selected. This resulted in 10,629 (out of 52,406) disease-causing amino acid substitutions and 8,417 (out of 282,625) unlabeled amino acid substitutions being successfully mapped to high-quality PDB structures. Table 1 summarizes both data sets. There exists an overlap between the HGMD (disease) and dbSNP (unlabeled) variants; the subset of dbSNP variants after the removal of HGMD variants will be referred to as putatively neutral variants (Table 1).

Functional site data sets

Metal ions annotated in X-ray structures from PDB as of May 2012 were selected using the HETATM field [52]. A metal-binding residue was defined as the residue that has at least one heavy atom (N, O or S) within $3^{\text{Å}}$ of the metal ion. In order to build an unbiased classifier, we removed chains with: (i) more than 40% sequence identity with any other chain in the data set, (ii) crystallographic resolution greater than or equal to 2.5Å, and (iii) R-values greater than or equal to 30%. We only considered data sets with more than 100 metal-binding residues and we refer to these residues as positives. N-linked glycosylation sites were parsed from PDB and filtered in the same way as the metal-binding sites. In the case of phosphorylation sites, we used the data set assembled previously [20]. Catalytic residues were collected from the Catalytic Site Atlas v2.2.10 [53] and only literature-supported sites were kept as positive examples. As with the previous data sets, we filtered out chains with more than 40% sequence identity. DNAbinding sites were collected from Yan et al. [54], RNA-binding sites were downloaded from ccPDB [55] and protein-protein interaction (PPI) sites were obtained from Chung et al. [56]. Each data set was further filtered to remove redundancy. Protein-protein interaction hot spot residues were obtained from Lise et al. [57]. This data set was created from ASEdb [58] and only sites that mapped to PDB were used. Chains with more than 35% sequence identity were filtered out in the original work. In our analysis, we used $\Delta\Delta G$ greater than 1kcal/mol as the

Category	Site type	nc	n ₊	AUC	sn	sp	мсс
Protein stability	Stability (S)	40	1041	0.735	0.118	0.989	0.223
Metal binding	Calcium (Ca)	1092	4860	0.895	0.448	0.986	0.561
	Cadmium (Cd)	199	1003	0.905	0.233	0.986	0.350
	Cobalt (Co)	156	532	0.934	0.596	0.986	0.623
	Copper (Cu)	105	440	0.951	0.780	0.985	0.727
	Iron (Fe)	187	785	0.976	0.875	0.986	0.843
	Potassium (K)	287	978	0.679	0.129	0.986	0.211
	Magnesium (Mg)	1348	3282	0.859	0.435	0.986	0.561
	Manganese (Mn)	366	1344	0.945	0.665	0.985	0.727
	Sodium (Na)	961	2753	0.671	0.105	0.987	0.211
	Nickel (Ni)	254	680	0.932	0.565	0.986	0.621
	Zinc (Zn)	1307	5778	0.966	0.623	0.987	0.691
PTMs	N-glycosylation (Nglyco)	339	736	0.785	0.120	0.986	0.183
	Phosphorylation (Phos)	655	1157	0.810	0.375	0.987	0.504
Catalytic activity	Catalytic (Cat)	721	2224	0.934	0.433	0.985	0.561
Macromolecular binding	DNA-binding (DNA)	139	3791	0.815	0.193	0.987	0.332
	RNA-binding (RNA)	83	3436	0.783	0.187	0.985	0.319
	Protein-protein interaction (PPI)	112	4350	0.807	0.091	0.987	0.191
	PPI hot spots (Hotspot)	35	165	0.803	0.309	0.986	0.278
Ligand binding	ADP	162	2589	0.842	0.335	0.985	0.475
	ATP	104	1733	0.813	0.242	0.986	0.382
	FAD	80	2248	0.840	0.307	0.985	0.448
	FMN	42	788	0.824	0.284	0.985	0.384
	GDP	45	593	0.843	0.433	0.985	0.502
	GTP	22	366	0.716	0.145	0.986	0.181
	НЕМ	83	2246	0.847	0.220	0.986	0.361
	NAD	73	1663	0.831	0.259	0.985	0.393
	PLP	34	477	0.916	0.505	0.986	0.543
	UDP	27	398	0.684	0.080	0.987	0.103
Allosteric regulation	Allosteric (Allo)	108	682	0.636	0.041	0.985	0.050

Table 2. Performance assessment of structural and functional residue predictors using cross-validation on positive-unlabeled data sets.

For each data set, we show the number of protein chains (n_c) and the number of positive examples (n_+). Additionally, we choose a score threshold corresponding to a specificity (sp) of 99% and report sensitivity (sn) and MCC at this threshold, as well as AUC. In each classification problem, the number of unlabeled examples was set to 10,000. <u>S1 Table</u> predictions lists the full name of each ligand code used. For the purposes of this work, structurally important amino acid residues such as specific metal ion binding residues were considered a part of the portfolio of available residue functions.

doi:10.1371/journal.pcbi.1005091.t002

cutoff for hot spots. Ligand binding data sets were collected from ccPDB as of November 2014 and allosteric site data were downloaded from ASD v2.0 [59]. For each data set, we removed chains with resolution greater than or equal to 2.5Å. Protein stability data was collected from Capriotti et al. [41]. We used the S1615 data set which consists of 1615 single site mutations extracted from 42 different proteins in the ProTherm database [60]. The attributes for each data point included solvent accessibility, pH value, temperature and energy change $\Delta\Delta G$. As previously noted, positive data points comprised mutations with $|\Delta\Delta G| > 0.5$. We further filtered out 112 redundant data points.

<u>Table 2</u> summarizes the protein stability and functional site data sets used in this study. In all situations, the unlabeled data set was constructed using a random sample of 10,000 residues selected from the 40% non-redundant set of monomers in PDB. This set was modified in the

case of post-translational modifications to include only modifiable residues; e.g., Asn for Nlinked glycosylation and Ser/Thr/Tyr for phosphorylation. It is important to emphasize that the set of negative examples was allowed to contain both buried and surface-exposed residues resulting in somewhat easier downstream classification problems. On the other hand, it allowed us to apply our methods to all PDB-mappable amino acid substitutions and make unbiased inferences related to different data sets.

Results

In this section we present the development of a stability model and a series of structure-based functional site predictors in order to examine the molecular effects of genetic variants. We evaluate the predictors through cross-validation and using an independent data set. We then summarize our results in relation to the functional impact of disease-causing substitutions and compare them to putatively neutral variants.

Assessment of functional site predictors

All classifiers developed in this study were constructed using positive and unlabeled data summarized in Table 2. Their performance was estimated via per-chain 10-fold cross-validation and is also shown in Table 2. S2 Table further lists the parameters for the best-performing kernel matrix obtained from a grid search over \mathcal{K} , $|\Sigma| = \{20, 40\}$, $m = \{0, 1\}$ and $N = \{4, 5\}$. Each predictor performance was assessed by means of the area under the ROC curve (AUC), sensitivity (*sn*) at 99% level of specificity (*sp*), and the Matthews correlation coefficient (MCC). The majority of predictors (26 out of 30) show good performance ($\geq 70\%$ AUC); however, we observe that functions related to smaller interfaces such as metal ions and active sites exhibit higher performance than other functional predictors. This result is not unexpected because predictors of macromolecular binding would have benefited from incorporating higher-order structural signatures such as clefts and pockets [20].

We also use an independent data set to evaluate a subset of functional site predictors, as depicted in <u>S3 Table</u>. Interestingly, most predictors, except for macromolecular binding models show similar or improved performance (AUC) values compared to those reported from cross-validation (<u>Table 2</u>). Overall, despite the variability of performance accuracies, limited number of independent data sets and the relatively small size of the validation data, these results provide evidence that functional site predictions are of sufficient quality to identify possible molecular alterations resulting from specific missense mutations.

A literature survey suggests that our predictors perform well when compared to established structure-based methods. Extensive comparisons with other work are difficult and were beyond the scope of this study as our main goal was to probabilistically assess molecular mechanisms of disease. A set of predictors built using the same methodology was best suited to this task.

Estimating prior and posterior probabilities

To use the formal framework laid out in the Methods section, it is important that all methods approximate posterior distributions. Using positive and unlabeled data, we have approached this problem in two steps: (i) by developing classifiers that discriminate between labeled and unlabeled data, and (ii) by estimating the class priors of the positive class in the unlabeled data [27].

Estimated class priors are a particularly useful by-product of learning posterior distributions. For the stability predictor, we estimate up to 13% of unlabeled variants to significantly impact stability using the AlphaMax algorithm [27]. When the stability model was applied to disease variants only, we estimate 14% of these variants to be impactful using the empirical mean formula. It should be noted that when the known disease variants were removed from the unlabeled data set, only 7% of the remaining variants were estimated to severely impact stability. In the case of functional predictors, we applied the AlphaMax algorithm using a set of positive variants and a set of 10,000 variants randomly sampled from a set of non-redundant monomers in PDB (S4 Table). In the case of catalytic residues, we estimate that up to 3% of PDB residues to be catalytic; however, about 5% of disease-causing and 2% of putatively neutral variants were estimated to be catalytic residues, etc. Overall, we generally observe a larger fraction of function-impacting variants in the disease-causing data set as compared with the putatively neutral variants.

Applying loss and gain functional site predictors to human variants

We applied the structure-based predictors on both the wild-type and mutant structural environments as a means to identify and categorize the functional effects of amino acid substitutions causing inherited disease. The distribution of scores on the putatively neutral variants was used as an empirical null distribution. We then used a particular false positive rate (FPR) value to determine a prediction threshold at which to assess the fraction of disease mutations with loss or gain scores that are as high as or higher than the threshold.

Table 3 summarizes the relative contributions of disease mutations that either decrease (loss) or increase (gain) the propensity of functional sites at a conservative threshold of 1% FPR for six different prediction outputs. Fig 2 visualizes a subset of these results for the case when stability is not impacted. Together with Table 3, it provides evidence that the loss and gain of functional sites exist even when protein stability is not disrupted; e.g., in the case of loss of function see columns $P(loss|\bar{S}, x)$ and $P(loss, \bar{S}|x)$ that roughly have the same values as P(loss|x). We mention that when either a loss or a gain of function event is found to be statistically significant, the mutation of this type of functional residue is considered to be an active mechanism of genetic disease. For instance, at 1% FPR, we observe that loss of catalytic residues (Cat; 3.34%; p-value = $1.93 \cdot 10^{-28}$) and iron-binding residues, (Fe; 3.17%; p-value = $2.06 \cdot 10^{-25}$) are among the most significantly affected molecular mechanisms.

<u>Table 3</u> also summarizes the statistical enrichment of impact on at least one functional site from the entire repertoire of functions using the independence and max models (see <u>Methods</u>). Here we observe a strong enrichment in all categories of loss of function, with or without impact on stability, for both the independence and max models. Additionally, we also see an enrichment in the gain-of-function events. These results provide statistical support for many individual studies that identify loss of function as a signature of human inherited disease.

Overall, our results suggest that with some exceptions, the loss of functional residues is enriched and common in human inherited disease; similarly, the gain of functional residues is observed to be an active mechanism in catalytic activity, most types of ligand-binding residues, and majority of metal-binding residues. In contrast to previous studies, our results suggest that the loss and gain of PTM sites do not show statistically significant enrichment in disease (although we observe enrichment for the loss); however, we note that this may be due to a considerable reduction of training data imposed by the availability of protein 3D structures, especially given a relationship between post-translational modifications and intrinsically disordered proteins [61-64].

Table 4 shows the proportions of disease and putatively neutral variants across functional categories for which molecular mechanisms can be computationally hypothesized. In the first part of the table, we compute the fraction of variants for which exactly one of the member predictors reports a score as high or higher than the FPR-value determined threshold. These fractions were then computed separately for disease and neutral variants. For convenience, when a

Data set		Single type loss events (%)			Single type gain events (%)			
	$P(\text{loss} \bar{S},x)$	$P(loss, \bar{S} x)$	P(oss x)	$P(gain \bar{S},x)$	$P(gain, \bar{S} x)$	P(gain x)		
Ca	2.65*	2.35*	2.69*	3.90*	3.63*	3.63*		
Cd	1.42	1.31	1.44	1.20	1.02	1.02		
Со	1.97*	1.94*	1.98*	0.80	0.70	0.70		
Cu	1.87*	1.82*	1.88*	1.29	1.15	1.15		
Fe	3.14*	2.97*	3.17*	1.77*	1.62*	1.62*		
К	2.45*	2.08*	2.45*	2.99*	2.60*	2.60*		
Mg	3.22*	2.83*	3.14*	3.09*	2.89*	2.89*		
Mn	2.42*	2.22*	2.45*	2.65*	2.47*	2.47*		
Na	3.10*	2.55*	3.13*	2.50*	3.05*	3.05*		
Ni	1.39	1.33	1.40	0.62	0.54	0.54		
Zn	2.86*	2.64*	2.89*	1.81*	1.63*	1.63*		
Nglyco	3.75	1.25	2.81	0.46	0.23	0.23		
Phos	1.69	1.47	1.69	0.58	0.46	0.46		
Cat	3.18*	2.90*	3.34*	4.45*	3.94*	3.94*		
DNA	1.58*	1.39	1.61*	1.85*	1.75*	1.75*		
RNA	0.96	0.89	0.96	1.25	1.05	1.05		
PPI	1.53*	1.27	1.65*	1.94*	1.79*	1.79*		
Hotspot	1.00	0.90	1.00	1.53*	1.45	1.45		
ADP	3.12*	2.92*	3.16*	4.03*	3.68*	3.68*		
ATP	2.73*	2.52*	2.76*	2.76*	2.41*	2.41*		
FAD	2.77*	2.58*	2.81*	3.21*	2.92*	2.92*		
FMN	2.15*	2.01*	2.17*	2.40*	2.18	2.18*		
GDP	2.07*	1.99*	2.08*	2.41*	2.18*	2.18*		
GTP	1.72*	1.48	1.72*	2.78*	2.15*	2.15*		
HEM	2.26*	2.05*	2.35*	2.39*	2.05*	2.05*		
NAD	3.00*	2.64*	3.09*	2.53*	2.15*	2.15*		
PLP	3.15*	2.98*	3.10*	2.61*	2.37*	2.37*		
UDP	2.70*	2.41*	2.71*	3.11*	2.46*	2.46*		
Allo	1.47	1.24	1.49	1.96*	1.74*	1.74*		
Model	Multi-typ	Multi-type loss events (%)		Multi-type gain events (%)				
	$P(\text{loss} \bar{S},x)$	$P(loss, \bar{S} x)$	P(oss x)	$P(gain \bar{S},x)$	$P(gain, \bar{S} x)$	P(gain x)		
Independence	3.71*	3.31*	3.89*	4.13*	3.37*	3.37*		
Max	3.50*	1.93*	3.56*	4.35*	2.63*	2.63*		

Table 3. Percentage of disease variants with prediction scores at the 1% false positive rate threshold in putatively neutral variants.

For each of the six prediction outputs and each function f, we show the percentage (%) of disease mutations that have a greater probability of loss and gain of function than a threshold corresponding to a 1% false positive rate (FPR). <u>S1</u> and <u>S2</u> Figs show an instance of the inverse cumulative distribution function of P(|oss|x) and P(gain|x), respectively. These thresholds were estimated from the empirical null distributions of the probability of loss or gain of function on the set of dbSNP neutral data.

*Indicates significant *p*-value measured by a one-tailed Fisher's exact test after Bonferroni correction for multiple comparisons. The *p*-value was separately estimated for each type of posterior distribution, jointly for loss and gain events ($p < \frac{0.05}{58} = 8.62 \cdot 10^{-4}$). The *p*-values for the combined models were corrected separately ($p < \frac{0.05}{4} = 1.25 \cdot 10^{-2}$).

doi:10.1371/journal.pcbi.1005091.t003

predictor outputs a value as high or higher than the value determined by a 1% FPR, we refer to this prediction as actionable hypothesis of loss or gain of function. On the other hand, when the FPR-based threshold is adjusted using the Bonferroni correction, we refer to these predictions are confident. For example, at a conservative *p*-value cutoff of $p < 8.62 \cdot 10^{-4}$, we find



Fig 2. Percentage of disease variants with prediction scores at the 1% false positive rate threshold in putatively neutral variants. For each function *f*, the bars indicate the percentage (%) of disease mutations that have a greater $P(|oss|\bar{S},x)$ and $P(gain|\bar{S},x)$ than a conservative threshold at 1% false positive rates (FPR). These thresholds are estimated from the null distributions of $P(|oss|\bar{S},x)$ and $P(gain|\bar{S},x)$ on the set of dbSNP neutral data, respectively. *Indicates significant *p*-value measured as a one-tailed Fisher's Exact test after Bonferroni correction for multiple hypothesis testing ($p < 8.62 \cdot 10^{-4}$). The red line indicates the percentage of neutral variants that have greater $P(|oss|\bar{S},x)$ and $P(gain|\bar{S},x)$ and $P(gain|\bar{S},x)$ which is exactly 1%.

doi:10.1371/journal.pcbi.1005091.g002

that 1.51% of mutations are likely to alter exactly one metal binding site and 1.43% may alter a single ligand binding site. For all groups of molecular mechanisms, we observe that the probability of observing a high alteration score is more than three times as likely as in the case of putatively neutral variants.

Table 4 also shows situations with two or more functional perturbations consequent to the replacement of a given amino acid residue. The amino acid substitutions disrupting multiple functions may be important in a therapeutic context because addressing a single deficiency (e.g. iron binding) may still not result in a fully corrected phenotype because other deficiencies may still remain (e.g. ligand binding). Here, we have a significantly increased likelihood of observing multi-functional alterations in the disease set compared to the putatively neutral set; i.e., the disease set is several times more likely to contain multi-functional alterations than the putatively neutral set. For instance, 1.23% of disease mutations are likely to affect at least two metal binding sites versus only 0.27% of neutral variants, whereas 0.68% of disease variants may affect more than one ligand binding site as opposed to 0.19% of neutral polymorphisms.

If we combine the results for single and multiple mechanisms, we observe that 2.24% of disease variants are predicted, with high confidence, to impair metal-binding sites (1.51% loss of single site and 0.72% loss of multiple sites) and 1.67% probably impair ligand binding sites (1.43% loss of single site and 0.24% loss of more than one site), as depicted in Fig 3. Overall, we believe we can confidently propose molecular mechanisms of disease for 8.6% of all variants in the inherited disease data set whereas we only see about 2.4% of such variants in the neutral

Category	Loss (%)		Gain (%)		Loss or Gain (%)				
	Disease	Neutral	Disease	Neutral	Disease	Neutral			
	I. Single mechanism								
Confident biological hypotheses (p -value < 8.62 \cdot 10 ⁻⁴)									
Metal binding	1.51	0.27	1.46	0.46	2.88	0.70			
PTMs	0	0	0.01	0	0.01	0			
Catalytic sites	0.73	0.06	0.40	0.07	1.14	0.14			
Macromolecule binding	0.59	0.20	0.64	0.20	1.21	0.40			
Ligand binding	1.43	0.47	1.86	0.46	3.07	0.87			
Allosteric sites	0.09	0.06	0.25	0.06	0.35	0.12			
All	3.03	0.84	3.29	0.97	5.80	1.70			
Actionable biological hypotheses (p-value < 0.01)									
Metal binding	4.90	2.50	5.08	2.99	8.70	5.03			
PTMs	0.30	0.17	0.08	0.22	0.39	0.40			
Catalytic sites	3.34	1.00	3.94	1.00	7.25	2.00			
Macromolecule binding	3.89	2.93	4.03	3.26	7.39	5.75			
Ligand binding	9.74	4.65	9.55	4.97	13.72	7.06			
Allosteric sites	1.49	1.00	1.74	1.00	3.16	2.00			
All	13.28	8.35	12.93	9.21	17.43	13.36			
II. Multiple mechanisms									
	Conf	ident biological hypo	otheses (p-value < 8.	62 · 10 ⁻⁴)					
Metal binding	0.72	0.16	0.49	0.10	1.23	0.27			
Macromolecule binding	0.05	0.04	0.17	0.04	0.22	0.07			
Ligand binding	0.24	0.07	0.33	0.09	0.68	0.19			
All	1.43	0.34	1.29	0.31	2.78	0.68			
	Ad	ctionable biological h	nypotheses (p-value	< 0.01)					
Metal binding	5.96	2.31	4.75	2.42	10.37	4.62			
Macromolecule binding	0.66	0.51	0.99	0.37	1.72	0.89			
Ligand binding	5.55	1.90	5.43	1.85	11.04	4.11			
All	12.88	5.42	12.22	5.42	23.57	10.60			

Table 4. Relative contribution of loss and gain of functional categories from amino acid substitutions.

For each functional site category, we show the relative contributions (%) of disease and neutral substitutions where at least one function *f* within a category has a greater P(|oss|x) or P(gain|x) than a conservative threshold at 1% FPR. This threshold is estimated from the null distributions of P(|oss|x) and P(gain|x) on the putatively neutral polymorphisms data set, respectively. The table is subdivided into two parts: (i) exactly one function (or mechanism) and (ii) two or more mechanisms. In both parts, the relative contributions are assessed at two p-value cutoffs of $p < 8.62 \cdot 10^{-4}$ and p < 0.01. Note that in a small number of cases, a loss of one function might result in the gain of another; thus, the sets of residues counted in the loss and gain may overlap.

doi:10.1371/journal.pcbi.1005091.t004

set. If we use a *p*-value cutoff of 0.01 without a Bonferroni correction, then we can computationally hypothesize a molecular mechanism for approximately 40.9% of disease variants.

Validation of loss of function predictions

In this study, we have proposed a novel methodology for identifying specific molecular alterations of disease mutations. Given that it is impractical to experimentally validate the predicted functional effects of each individual amino acid substitution, we use mutagenesis experimental data to independently assess the loss of functional site predictions, as shown in <u>S5 Table</u>. To the best of our knowledge, this is the first time a systematic assessment of computationally predicted disruptions of specific types of functional residues has been carried out in the published literature.



Fig 3. Relative contribution of loss and gain of functional categories on each amino acid substitutions data set. For each functional site category, we show the relative contributions (%) of disease and neutral variants where at least one function *f* within a category has a greater P(|oss|x) or P(gain|x) than a conservative threshold at 1% FPR. This threshold is estimated from the null distribution of P(|oss|x) and P(gain|x) on the putatively neutral polymorphisms data set, respectively. *Indicates significant *p*-value measured as a Fisher's Exact test after Bonferroni correction for multiple hypothesis comparisons ($p < 8.62 \cdot 10^{-4}$).

doi:10.1371/journal.pcbi.1005091.g003

In general, our loss of function predictors performed as expected. However, more interestingly, if one restricts the loss of function predictions to those with significant *p*-values (i.e. p < 0.01), then performance (AUC) rises to at least 95% for all predictors. This provides compelling evidence that our methodology can be effectively used to identify molecular mechanisms of disease and hence can be used to prioritize experimental validation. Additionally, Fig 4 depicts two case studies of loss and gain of function predictions which have been experimentally validated. We discuss each case in detail below:



Fig 4. 3D visualization of protein structures with experimentally supported loss and gain of function predictions. Left: SOD1 protein (chain A of PDB entry 2xjl) where residues H63, H71, H80 and D83 form a zinc binding pocket. The substitution D83G gives rise to a loss of zinc binding. Right: CA2 protein (chain A of PDB entry 1fqr) where H94, H96 and H119 are zinc-binding sites. Mutation T198E leads to an increase in zinc affinity.

doi:10.1371/journal.pcbi.1005091.g004

Loss of zinc binding in superoxide dismutase (SOD1). The functional role of SOD1 is to destroy radicals that are normally produced in cells and which are toxic to biological systems. SOD1 forms a zinc-binding pocket consisting of H63, H71, H80 and D83 [65, 66] as shown in Fig 4 (left). Mutations in SOD1 are known to be causative of amyotrophic lateral sclerosis [66–68]. However, the molecular mechanisms underlying these mutations often remain unclear. We predicted a loss of multiple functional activities for mutation D83G and identified zinc binding as the primary underlying molecular mechanism of disease. In particular, D83G has a $P(f|x'_{wt}) = 0.99$ and $P(f|x'_{mt}) = 3.3 \cdot 10^{-3}$ leading to a $P(\log |x|) \approx 1$, which is above the 1% FPR threshold of 0.20 with an empirical *p*-value of $1.2 \cdot 10^{-3}$. A literature search for experimental evidence reveals that mutation D83G causes the destabilization of native structure which leads to protein aggregation with the formation of amyloid-like fibrils, and, ultimately, a gain of toxicity [69]. Zinc binding is a known stabilizer of protein structure and, therefore, the loss of the zinc-binding residue D83 appears to be a plausible destabilizing mechanism that ultimately impacts the biological function of SOD1. We note that the quadruple (H63, H71, H80, D83) was not part of the training data for the zinc-binding predictor.

This example raises an interesting possibility that the loss of a functionally important residue (zinc-binding residue) results in a loss of stability, and ultimately leads to disease through the loss of the protein's function. In other words, protein structure and function appears to be intimately and bidirectionally interconnected. At this moment, however, this is only a theoretical possibility because of the lack of data about the structure and stability of the wild-type and mutant proteins in the absence of zinc ions.

Gain of zinc affinity in carbonic anhydrase 2 (CA2). CA2 is essential for bone resorption and osteoclast differentiation. CA2 has three zinc-binding residues at H94, H96 and H119 as shown in Fig 4 (right). There are multiple studies that have characterized the effects of variants in CA2 via mutagenesis experiments [70–74]. Among these mutations, we predicted a gain of zinc binding for T198E that was experimentally shown to increase zinc affinity. Specifically, T198E has a $P(f|x'_{wt}) = 3.8 \cdot 10^{-4}$ and $P(f|x'_{mt}) = 0.99$ leading to a $P(gain|x) \approx 1$, which is above the 1% FPR threshold of 0.35 with a *p*-value of $3.7 \cdot 10^{-4}$. The triple (H94, H96, H119) was not part of the training data for the zinc-binding residue predictor.

Discussion

This study builds on the extensive prior work in structural bioinformatics to provide statistical evidence of the important role that alterations of multiple types of functional residue play in human genetic disease. Most of the existing work has centered around understanding the impact of sequence variants on protein stability or has only considered single types of function such as catalytic residues or protein-interaction sites [7, 20, 23, 24, 30, 75–78]. This work extends these studies by integrating the stability models with a series of functional residue predictors involving metal binding, macromolecular binding, ligand binding and others. Overall, we show and validate the feasibility of computationally predicting mutations that impair specific function using protein 3D structure data.

Despite using sophisticated methodology to model loss and gain of functional residues, the nature of this research has limitations involving both data sets and methodology. First, despite major efforts employed by authors and database curators when annotating amino acid substitutions as being causative of a particular disease, it is possible that some amino acid substitutions have been misannotated as disease-causing by the original authors reporting them. Similarly, mutagenesis experimental data are known to be biased toward certain amino acid residues. For example, alanine mutations comprised about 50% of the independent amino acid substitutions data set (due to the frequent use of alanine-scanning mutagenesis). There are also limitations and biases in relation to the protein structures available in PDB as well as in selecting an appropriate set of unlabeled variants. Second, there exist both theoretical and practical limitations in the semi-supervised framework used in this work. The accuracy of our methods is predicated upon the assumption that the computational models are capable of accurately estimating the posterior probability of the class labels. This however could not be guaranteed and thus requires caution when interpreting our results. Furthermore, there are identifiability issues in estimating class priors in the positiveunlabeled framework; i.e., the estimates for the class priors do not have a unique solution and only an upper bound can be estimated [27]. On the practical side, we have been careful to prevent overfitting. We performed only minor parameter selection steps before the final functional predictors were built. Thus, there is the potential to further improve predictor performance through more extensive work. This includes the use of additional features, optimizing the distance threshold used to define an edge between two residues when constructing protein contact graphs, choice of the capacity parameter in SVM^{light}, among others. Finally, this work was designed to probabilistically reason about molecular mechanisms of disease and not necessarily to develop classifiers that outperform specialized models across the board. If a user needs a tool for a particular prediction task, we recommend that the most accurate predictor for this task be selected.

Despite these limitations, we believe this work contributes to an improved understanding of the impact of sequence variants on protein function. We have provided a model that considers functional alteration both when stability of the protein is disrupted and when it is not disrupted (e.g. interestingly, sequence changes can exert a functional effect in disordered regions such as disorder-to-order transition [79]). We believe that our work suggests a new class of approaches to disease studies that might qualify as mechanism-driven and disease-agnostic, where one might be compelled to identify a set of molecular alterations underlying a disease phenotype without necessarily studying a single disease. While each molecular alteration is likely to require an individualized approach to drug design and therapy, we envisage that the next generation of researchers might decide to specialize in addressing particular types of functional deficiencies rather than beginning with a particular disease.

Supporting Information

S1 Fig. Inverse cumulative distribution function (CDF) of P(loss|x). For ATP-binding predictor, we plot the inverse CDF for P(loss|x) on the disease and putatively neutral data sets, respectively.

(EPS)

S2 Fig. Inverse cumulative distribution function (CDF) of P(gain|x). For catalytic residue predictor, we plot the inverse CDF for P(gain|x) on the disease and putatively neutral data sets, respectively.

(EPS)

S1 Table. Mapping between ligand codes and names. (PDF)

S2 Table. Selected kernel matrix parameters for each structural and functional site predictor. For each data set, we show the best-performing kernel matrix parameters obtained through a per-chain 10-fold cross-validation. The normalized edit distance kernel $k_m(u, v)$ outperformed both $k_m^l(u, v)$ and $k_m^e(u, v)$ on each data set. Note that the edit distance kernel with m = 0 is equivalent to a standard graphlet kernel. (PDF) S3 Table. Performance assessment of functional residue predictors using an independent data set. For each prediction method, we show number of proteins (n_p) , number of positive examples (n_+) , number of unlabeled examples (n_u) , AUC and sensitivity (sn) and MCC at score threshold corresponding to specificity (sp) of 99%. In the case of N-linked glycosylation (Nglyco), we only predict if the wild-type residue is an asparagine, whereas for phosphorylation (Phos), we only make predictions on threonine, tyrosine or serine residues. (PDF)

S4 Table. Class priors for structural and functional predictors. Fraction of residues in a data set estimated to be stability-impacting or functional. Estimates on the unlabeled data were made using the AlphaMax algorithm [27]; minor manual adjustments were made by observing the log-likelihood plots. *Indicates a confident prior estimate assessed by manually observing log-likelihood plots. Estimates on the disease and putatively neutral data were made using the empirical mean formula.

(PDF)

S5 Table. Performance assessment of loss of function predictions using mutagenesis experimental data. Independent validation of predicted loss of functional site events using a set of mutagenesis experimental data mapped to protein structures in PDB. This mutagenesis data set contains 3,356 AAS from 880 human proteins. For each functional feature, we show the number of experimentally determined losses (n_i), AUC, sensitivity (sn) and MCC corresponding to a 99% specificity (sp) threshold. Additionally, the last five columns show the number of statistically significant (p < 0.01) loss-of-function predictions (n_i^*), as well as estimates for AUC (AUC^{*}), sensitivity (sn^*), specificity (sp^*) and MCC (MCC^{*}) on this filtered set. (PDF)

Acknowledgments

We thank Prof. Charles Dann III for helpful comments.

Author Contributions

Conceived and designed the experiments: JLM PR SDM.

Performed the experiments: JLM.

Analyzed the data: JLM VP KAP SJ MM DNC SDM PR.

Wrote the paper: JLM MM DNC SDM PR.

References

- Sherry ST, Ward MH, Kholodov M, Baker J, Phan L, Smigielski EM, et al. dbSNP: the NCBI database of genetic variation. Nucleic Acids Res. 2001; 29(1):308–311. doi: <u>10.1093/nar/29.1.308</u> PMID: <u>11125122</u>
- Bairoch A, Apweiler R, Wu CH, Barker WC, Boeckmann B, Ferro S, et al. The Universal Protein Resource (UniProt). Nucleic Acids Res. 2005; 33(Database issue):D154–159. doi: <u>10.1093/nar/gki070</u> PMID: <u>15608167</u>
- Stenson PD, Mort M, Ball EV, Shaw K, Phillips AD, Cooper DN. The Human Gene Mutation Database: building a comprehensive mutation repository for clinical and molecular genetics, diagnostic testing and personalized genomic medicine. Hum Genet. 2014; 133(1):1–9. doi: <u>10.1007/s00439-013-1358-4</u> PMID: <u>24077912</u>
- 4. Mooney S. Bioinformatics approaches and resources for single nucleotide polymorphism functional analysis. Brief Bioinform. 2005; 6(1):44–56. doi: <u>10.1093/bib/6.1.44</u> PMID: <u>15826356</u>

- Ng PC, Henikoff S. Predicting the effects of amino acid substitutions on protein function. Annu Rev Genomics Hum Genet. 2006; 7(1):61–80. doi: <u>10.1146/annurev.genom.7.080505.115630</u> PMID: <u>16824020</u>
- Karchin R. Next generation tools for the annotation of human SNPs. Brief Bioinform. 2009; 10(1):35– 52. doi: <u>10.1093/bib/bbn047</u> PMID: <u>19181721</u>
- Wang Z, Moult J. SNPs, protein structure, and disease. Hum Mutat. 2001; 17(4):263–270. doi: <u>10.</u> <u>1002/humu.22</u> PMID: <u>11295823</u>
- 8. Yue P, Li Z, Moult J. Loss of protein structure stability as a major causative factor in monogenic disease. J Mol Biol. 2005; 353(2):459–473. doi: <u>10.1016/j.jmb.2005.08.020</u> PMID: <u>16169011</u>
- Vogt G, Chapgier A, Yang K, Chuzhanova N, Feinberg J, Fieschi C, et al. Gains of glycosylation comprise an unexpectedly large group of pathogenic mutations. Nat Genet. 2005; 37(7):692–700. doi: <u>10.</u> 1038/ng1581 PMID: 15924140
- Vogt G, Vogt B, N C, Julenius K, Cooper DN, Casanova JL. Gain-of-glycosylation mutations. Curr Opin Genet Dev. 2007; 17(3):245–251. doi: 10.1016/j.gde.2007.04.008 PMID: 17467977
- Li S, lakoucheva L, Mooney SD, Radivojac P. Loss of post-translational modification sites in disease. Pac Symp Biocomput. 2010; 15:337–347. PMID: <u>19908386</u>
- Pan Y, Karagiannis K, Zhang H, Dingerdissen H, Shamsaddini A, Wan Q, et al. Human germline and pan-cancer variomes and their distinct functional profiles. Nucleic Acids Res. 2014; 42(18):11570– 11588. doi: <u>10.1093/nar/gku772</u> PMID: <u>25232094</u>
- Peterson TA, Doughty E, Kann MG. Towards precision medicine: advances in computational approaches for the analysis of human variants. J Mol Biol. 2013; 425(21):4047–4063. doi: <u>10.1016/j.</u> jmb.2013.08.008 PMID: <u>23962656</u>
- Radivojac P, Baenziger PH, Kann MG, Mort ME, Hahn MW, Mooney SD. Gain and loss of phosphorylation sites in human cancer. Bioinformatics. 2008; 24(16):i241–247. doi: <u>10.1093/bioinformatics/btn267</u> PMID: 18689832
- Radivojac P, Vacic V, Haynes C, Cocklin RR, Mohan A, Heyen JW, et al. Identification, analysis, and prediction of protein ubiquitination sites. Proteins. 2010; 78(2):365–380. doi: <u>10.1002/prot.22555</u> PMID: <u>19722269</u>
- Xin F, Myers S, Li YF, Cooper DN, Mooney SD, Radivojac P. Structure-based kernels for the prediction of catalytic residues and their involvement in human inherited disease. Bioinformatics. 2010; 26 (16):1975–1982. doi: <u>10.1093/bioinformatics/btq319</u> PMID: <u>20551136</u>
- Li B, Krishnan VG, Mort ME, Xin F, Kamati KK, Cooper DN, et al. Automated inference of molecular mechanisms of disease from amino acid substitutions. Bioinformatics. 2009; 25(21):2744–2750. doi: 10.1093/bioinformatics/btp528 PMID: 19734154
- Saunders CT, Baker D. Evaluation of structural and evolutionary contributions to deleterious mutation prediction. J Mol Biol. 2002; 322(4):891–901. doi: <u>10.1016/S0022-2836(02)00813-6</u> PMID: <u>12270722</u>
- Capriotti E, Altman RB. Improving the prediction of disease-related variants using protein three-dimensional structure. BMC Bioinformatics. 2011; 12 Suppl 4:S3. doi: <u>10.1186/1471-2105-12-S4-S3</u> PMID: 21992054
- Xin F, Radivojac P. Computational methods for identification of functional residues in protein structures. Curr Protein Pept Sci. 2011; 12:456–469. doi: <u>10.2174/138920311796957685</u> PMID: <u>21787297</u>
- Sahni N, Yi S, Taipale M, Fuxman Bass JI, Coulombe-Huntington J, Yang F, et al. Widespread macromolecular interaction perturbations in human genetic disorders. Cell. 2015; 161(3):647–660. doi: <u>10.</u> <u>1016/j.cell.2015.04.013</u> PMID: <u>25910212</u>
- 22. Steward RE, MacArthur MW, Laskowski RA, Thornton JM. Molecular basis of inherited diseases: a structural perspective. Trends Genet. 2003; 19(9):505–513. doi: <u>10.1016/S0168-9525(03)00195-1</u> PMID: <u>12957544</u>
- De Baets G, Van Durme J, Reumers J, Maurer-Stroh S, Vanhee P, Dopazo J, et al. SNPeffect 4.0: online prediction of molecular and structural effects of protein-coding variants. Nucleic Acids Res. 2012; 40(Database issue):D935–939. doi: 10.1093/nar/gkr996 PMID: 22075996
- Yates CM, Sternberg MJ. The effects of non-synonymous single nucleotide polymorphisms (nsSNPs) on protein-protein interactions. J Mol Biol. 2013; 425(21):3949–3963. doi: <u>10.1016/j.jmb.2013.07.012</u> PMID: 23867278
- Nussinov R, Tsai CJ, Xin F, Radivojac P. Allosteric post-translational modification codes. Trends Biochem Sci. 2012; 37(10):447–455. doi: 10.1016/j.tibs.2012.07.001 PMID: 22884395
- 26. Elkan C, Noto K. Learning classifiers from only positive and unlabeled data. In: Proceedings of the 14th ACM SIGKDD International Conference on Knowledge Discovery and Data Mining. KDD 2008. New York, NY, USA: ACM; 2008. p. 213–220.

- Jain S, White M, Trosset MW, Radivojac P. Nonparametric semi-supervised learning of class proportions. arXiv preprint arXiv:160101944. 2016;Available from: http://arxiv.org/abs/1601.01944.
- Denis F, Gilleron R, Letouzey F. Learning from positive and unlabeled examples. Theor Comput Sci. 2005; 348(16):70–83. doi: <u>10.1016/j.tcs.2005.09.007</u>
- Lagace TA, Curtis DE, Garuti R, McNutt MC, Park SW, Prather HB, et al. Secreted PCSK9 decreases the number of LDL receptors in hepatocytes and in livers of parabiotic mice. J Clin Invest. 2006; 116 (11):2995–3005. doi: <u>10.1172/JCI29383</u> PMID: <u>17080197</u>
- Bromberg Y, Rost B. Correlating protein function and stability through the analysis of single amino acid substitutions. BMC Bioinformatics. 2009; 10 Suppl 8:S8. doi: <u>10.1186/1471-2105-10-S8-S8</u> PMID: <u>19758472</u>
- Tokuriki N, Tawfik DS. Stability effects of mutations and protein evolvability. Curr Opin Struct Biol. 2009; 19(5):596–604. doi: <u>10.1016/j.sbi.2009.08.003</u> PMID: <u>19765975</u>
- Worth CL, Preissner R, Blundell TL. SDM–a server for predicting effects of mutations on protein stability and malfunction. Nucleic Acids Res. 2011; 39(Web Server issue):W215–222. doi: <u>10.1093/nar/gkr363</u> PMID: <u>21593128</u>
- du Plessis MC, Sugiyama M. Class prior estimation from positive and unlabeled data. IEICE Transactions on Information and Systems. E97–D(5):1358–1362.
- Przulj N, Corneil DG, Jurisica I. Modeling interactome: scale-free or geometric? Bioinformatics. 2004; 20(18):3508–3515. doi: <u>10.1093/bioinformatics/bth436</u> PMID: <u>15284103</u>
- Przulj N. Biological network comparison using graphlet degree distribution. Bioinformatics. 2007; 23(2): e177–e183. doi: 10.1093/bioinformatics/btl301 PMID: 17237089
- Shervashidze N, Vishwanathan SVN, Petri TH, Mehlhorn K, Borgwardt KM. Efficient graphlet kernels for large graph comparison. In: Proc. 12th International Conference on Artificial Intelligence and Statistics. AISTATS'09; 2009. p. 488–495.
- Vacic V, lakoucheva LM, Lonardi S, Radivojac P. Graphlet kernels for prediction of functional residues in protein structures. J Comput Biol. 2010; 17(1):55–72. doi: 10.1089/cmb.2009.0029 PMID: 20078397
- Lugo-Martinez J, Radivojac P. Generalized graphlet kernels for probabilistic inference in sparse graphs. Network Science. 2014; 2(2):254–276. doi: <u>10.1017/nws.2014.14</u>
- Joachims T. Learning to classify text using support vector machines: methods, theory, and algorithms. Kluwer Academic Publishers; 2002.
- Platt J. Probabilistic outputs for support vector machines and comparisons to regularized likelihood methods. In: Advances in Large Margin Classifiers. MIT Press; 1999. p. 61–74.
- Capriotti E, Fariselli P, Casadio R. A neural-network-based method for predicting protein stability changes upon single point mutations. Bioinformatics. 2004; 20(1):i63–i68. doi: <u>10.1093/bioinformatics/</u> <u>bth928</u> PMID: <u>15262782</u>
- Cheng J, Randall A, Baldi P. Prediction of protein stability changes for single-site mutations using support vector machines. Proteins. 2006; 62(4):1125–1132. doi: 10.1002/prot.20810
- Kyte J, Doolittle RF. A simple method for displaying the hydropathic character of a protein. J Mol Biol. 1982; 157(1):105–132. doi: <u>10.1016/0022-2836(82)90515-0</u> PMID: <u>7108955</u>
- Janin J. Surface and inside volumes in globular proteins. Nature. 1979; 277(5696):491–492. doi: <u>10.</u> <u>1038/277491a0</u> PMID: <u>763335</u>
- Vihinen M, Torkkila E, Riikonen P. Accuracy of protein flexibility predictions. Proteins. 1994; 19(2):141– 149. doi: <u>10.1002/prot.340190207</u> PMID: <u>8090708</u>
- Jones S, Thornton JM. Principles of protein-protein interactions. Proc Natl Acad Sci USA. 1996; 93 (1):13–20. doi: <u>10.1073/pnas.93.1.13</u> PMID: <u>8552589</u>
- Jones S, Thornton JM. Analysis of protein-protein interaction sites using surface patches. J Mol Biol. 1997; 272(1):121–132. doi: <u>10.1006/jmbi.1997.1234</u> PMID: <u>9299342</u>
- Zimmerman JM, Eliezer N, Simha R. The characterization of amino acid sequences in proteins by statistical methods. J Theor Biol. 1968; 21(2):170–201. doi: <u>10.1016/0022-5193(68)90069-6</u> PMID: <u>5700434</u>
- George RA, Heringa J. An analysis of protein domain linkers: their classification and role in protein folding. Protein Eng. 2002; 15(11):871–879. doi: 10.1093/protein/15.11.871 PMID: 12538906
- Nagano K. Logical analysis of the mechanism of protein folding. I. Predictions of helices, loops and beta-structures from primary structure. J Mol Biol. 1973; 75(2):401–420. doi: <u>10.1016/0022-2836(73)</u> <u>90030-2</u> PMID: <u>4728695</u>
- Altschul SF, Madden TL, Schäffer AA, Zhang J, Zhang Z, Miller W, et al. Gapped BLAST and PSI-BLAST: a new generation of protein database search programs. Nucleic Acids Res. 1997; 25 (17):3389–3402. doi: <u>10.1093/nar/25.17.3389</u> PMID: <u>9254694</u>

- Xin F. Methods for predicting functional residues in protein structures and understanding molecular mechanisms of disease [Ph.D. Thesis]; 2012.
- Porter CT, Bartlett GJ, Thornton JM. The Catalytic Site Atlas: a resource of catalytic sites and residues identified in enzymes using structural data. Nucleic Acids Res. 2004; 32:D129–33. doi: <u>10.1093/nar/</u> <u>gkh028</u> PMID: <u>14681376</u>
- Yan C, Terribilini M, Wu F, Jernigan RL, Dobbs D, Honavar V. Predicting DNA-binding sites of proteins from amino acid sequence. BMC Bioinformatics. 2006; 7:262–271. doi: <u>10.1186/1471-2105-7-262</u> PMID: <u>16712732</u>
- Singh H, Chauhan JS, Gromiha MM, Consortium OSDD, Raghava GPS. ccPDB: compilation and creation of data sets from Protein Data Bank. Nucleic Acids Res. 2012; 40(D1):D486–D489. doi: <u>10.1093/</u> <u>nar/gkr1150</u> PMID: 22139939
- Chung JL, Wang W, Bourne PE. Exploiting sequence and structure homologs to identify protein–protein binding sites. Proteins. 2006; 62(3):630–640. doi: <u>10.1002/prot.20741</u> PMID: <u>16329107</u>
- Lise S, Archambeau C, Pontil M, Jones D. Prediction of hot spot residues at protein-protein interfaces by combining machine learning and energy-based methods. BMC Bioinformatics. 2009; 10(1):365– 381. doi: 10.1186/1471-2105-10-365 PMID: 19878545
- Thorn KS, Bogan AA. ASEdb: a database of alanine mutations and their effects on the free energy of binding in protein interactions. Bioinformatics. 2001; 17(3):284–285. doi: <u>10.1093/bioinformatics/17.3.</u> 284 PMID: <u>11294795</u>
- 59. Huang Z, Mou L, Shen Q, Lu S, Li C, Liu X, et al. ASD v2.0: updated content and novel features focusing on allosteric regulation. Nucleic Acids Res. 2014; 42(D1):D510–D516. doi: <u>10.1093/nar/gkt1247</u> PMID: 24293647
- Gromiha M, An J, Kono H, Oobatake M, Uedaira H, Prabakaran P, et al. ProTherm, version 2.0: thermodynamic database for proteins and mutants. Nucleic Acids Research. 2000; 28(1):283–285. doi: <u>10</u>. 1093/nar/28.1.283 PMID: 10592247
- Iakoucheva LM, Radivojac P, Brown CJ, O'Connor TR, Sikes JG, Obradovic Z, et al. The importance of intrinsic disorder for protein phosphorylation. Nucleic Acids Res. 2004; 32(3):1037–1049. doi: <u>10.1093/</u> nar/gkh253 PMID: 14960716
- Daily KM, Radivojac P, Dunker AK. Intrinsic disorder and protein modifications: building an SVM predictor for methylation. In: Proceedings of the IEEE Symposium on Computational Intelligence in Bioinformatics and Computational Biology (CIBCB). IEEE; 2005. p. 475–481.
- Gsponer J, Futschik ME, Teichmann SA, Babu MM. Tight regulation of unstructured proteins: from transcript synthesis to protein degradation. Science. 2008; 322(5906):1365–1368. doi: <u>10.1126/science.</u> <u>1163581</u> PMID: <u>19039133</u>
- Pejaver V, Hsu WL, Xin F, Dunker AK, Uversky VN, Radivojac P. The structural and functional signatures of proteins that undergo multiple events of post-translational modification. Protein Sci. 2014; 23 (8):1077–1093. doi: 10.1002/pro.2494 PMID: 24888500
- Roberts BR, Tainer JA, Getzoff ED, Malencik DA, Anderson SR, Bomben VC, et al. Structural characterization of zinc-deficient human superoxide dismutase and implications for ALS. J Mol Biol. 2007; 373 (4):877–890. doi: 10.1016/j.jmb.2007.07.043 PMID: 17888947
- 66. Seetharaman SV, Taylor AB, Holloway S, Hart PJ. Structures of mouse SOD1 and human/mouse SOD1 chimeras. Arch Biochem Biophys. 2010; 503(2):183–190. doi: <u>10.1016/j.abb.2010.08.014</u> PMID: <u>20727846</u>
- Alexander MD, Traynor BJ, Miller N, Corr B, Frost E, McQuaid S, et al. "True" sporadic ALS associated with a novel SOD-1 mutation. Ann Neurol. 2002; 52(5):680–683. doi: <u>10.1002/ana.10369</u> PMID: <u>12402272</u>
- Krishnan U, Son M, Rajendran B, Elliott JL. Novel mutations that enhance or repress the aggregation potential of SOD1. Mol Cell Biochem. 2006; 287(1–2):201–211. doi: <u>10.1007/s11010-005-9112-4</u> PMID: <u>16583143</u>
- Joyce PI, Mcgoldrick P, Saccon RA, Weber W, Fratta P, West SJ, et al. A novel SOD1-ALS mutation separates central and peripheral effects of mutant SOD1 toxicity. Hum Mol Genet. 2015; 24(7):1883– 1897. doi: <u>10.1093/hmg/ddu605</u> PMID: <u>25468678</u>
- Ippolito JA, Christianson DW. Structure of an engineered His3 Cys zinc binding site in human carbonic anhydrase II. Biochemistry. 1993; 32(38):9901–9905. doi: 10.1021/bi00089a005 PMID: 8399159
- Krebs JF, Ippolito JA, Christianson DW, Fierke CA. Structural and functional importance of a conserved hydrogen bond network in human carbonic anhydrase II*. J Biol Chem. 1993; 268(36):27458–27466. PMID: <u>8262987</u>

- 72. Xue Y, Liljas A, Jonsson BH, Lindskog S. Structural analysis of the zinc hydroxide-Thr-199-Glu-106 hydrogen-bond network in human carbonic anhydrase II. Proteins. 1993; 17(1):93–106. doi: <u>10.1002/prot.340170112</u> PMID: 7901850
- Ippolito JA, Baird TT Jr, M SA, C DW, F CA. Structure-assisted redesign of a protein-zinc-binding site with femtomolar affinity. Proc Natl Acad Sci USA. 1995; 92:5017–5021. doi: <u>10.1073/pnas.92.11.5017</u> PMID: <u>7761440</u>
- 74. Huang S, Sjöblom, Sauer-Eriksson AE, Jonsson BH. Organization of an efficient carbonic anhydrase: Implications for the mechanism based on structure-function studies of a T199P/C206S mutant. Biochemistry. 2002; 41(24):7628–7635. doi: 10.1021/bi0200530 PMID: 12056894
- 75. DePristo MA, Weinreich DM, Hartl DL. Missense meanderings in sequence space: a biophysical view of protein evolution. Nat Rev Genet. 2005; 6(9):678–687. doi: <u>10.1038/nrg1672</u> PMID: <u>16074985</u>
- 76. Schymkowitz J, Borg J, Stricher F, Nys R, Rousseau F, Serrano L. The FoldX web server: an online force field. Nucleic Acids Res. 2005; 33(Web Server issue):W382–388. doi: <u>10.1093/nar/gki387</u> PMID: <u>15980494</u>
- Casadio R, Vassura M, Tiwari S, Fariselli P, Luigi Martelli P. Correlating disease-related mutations to their effect on protein stability: a large-scale analysis of the human proteome. Hum Mutat. 2011; 32 (10):1161–1170. doi: 10.1002/humu.21555 PMID: 21853506
- 78. Gray VE, Kukurba KR, Kumar S. Performance of computational tools in evaluating the functional impact of laboratory-induced amino acid mutations. Bioinformatics. 2012; 28(16):2093–2096. doi: <u>10.1093/</u> <u>bioinformatics/bts336</u> PMID: <u>22685075</u>
- Mohan A, Uversky VN, Radivojac P. Influence of sequence changes and environment on intrinsically disordered proteins. PLoS Comput Biol. 2009; 5(9):e1000497. doi: <u>10.1371/journal.pcbi.1000497</u> PMID: <u>19730682</u>